Author's response to reviews

Title: Comparison of ICD code-based diagnosis of obesity with measured obesity in children and the implications for health care cost estimates

Authors:

Stefan Kuhle (stefan.kuhle@dal.ca)
Sara FL Kirk (sara.kirk@dal.ca)
Arto Ohinmaa (arto.ohinmaa@ualberta.ca)
Paul J Veugelers (paul.veugelers@ualberta.ca)

Version: 4 Date: 19 December 2011

Author's response to reviews:

To Whom It May Concern
Full author names in the "Contributions" section have been changed to initials.
The manuscript has been checked again for errors.
Thanks,
Stefan Kuhle